
diff -r 94988aa97974 -r 11f5f86bf956 handouts/ho01.tex
--- a/handouts/ho01.tex	Wed Aug 19 13:01:53 2015 +0800
+++ b/handouts/ho01.tex	Wed Sep 09 11:08:54 2015 +0100
@@ -597,7 +597,7 @@
 
 \subsubsection*{Further Reading}
 
-A readable article by Bruce Schneier on ``How Security Companies Sucker Us With 
+A readable article by Bruce Schneier on ``How Security Companies Sucker Us with 
 Lemons''
 
 \begin{center}
@@ -622,6 +622,13 @@
 \url{http://www.nostarch.com/pentesting}
 \end{center}
 
+\noindent Even the government recently handed out some 
+advice about passwords
+
+\begin{center}
+\url{http://goo.gl/dIzqMg}
+\end{center}
+
 \noindent Clearly, passwords are a technology that comes to
 the end of its usefulness, because brute force attacks become
 more and more powerful and it is unlikely that humans get any




